%&LaTeX
\documentclass{article}

\usepackage[utf8]{inputenc}
\usepackage[T1]{fontenc}
\usepackage{textcomp}

\begin{document}

\section*{2023}

Suarez Patricia, Carpio Dario, \& Sappa Angel D. (2023). A Deep Learning Based Approach for Synthesizing Realistic Depth Maps. In \textit{Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics 22nd International Conference on Image Analysis and Processing, ICIAP 2023 Udine 11 -- 15 September 2023} (Vol. 14234 LNCS, pp. 369--380).

\end{document}

